Author’s response to reviews

Title: Duplication of Superficial Femoral Artery: Imaging Findings and Literature Review

Authors:

Sung Hyun Yu (ysh4627@naver.com)
Jung Han Hwang (lj3800@naver.com)
Jeong Ho Kim (ho7ok7@gilhospital.com)
Suyoung Park (hwp20@naver.com)
Ki Hyun Lee (singbaram@naver.com)
Sang Tae Choi (bcon1218@naver.com)

Version: 1 Date: 11 Aug 2020

Author’s response to reviews:

Point-by-point response

Journal Requirements:

1. Please consider the list of authors as it currently stands with reference to our guidelines regarding qualification for authorship. Currently, the contributions of authors YSH and CST do not automatically qualify them for authorship. In the section “Authors’ contributions”, please provide further clarifications on their contributions, and see our guidelines for authorship.

Response) We have understood your explanation, and corrected the authorship according to journal’s guidelines.

Authors’ contributions
YSH: Acquisition and analysis of the work, Drafted the work
HJH: Conception of the work and substantively revised it
KJH: Design of the work and substantively revised it
LKH: Writing – review & editing
PS: Writing – review & editing
CST: Writing – review & editing
All authors have checked the authorship to a submitted version and agreed to the author list and contributions.